Author's response to reviews

**Title:** HIV associated hypocalcaemia among diarrheic patients in northwest Ethiopia: A cross sectional study

**Authors:**

Beyene Moges (beyemoges@gmail.com)
Bemnet Amare (amarebem6@gmail.com)
Timoki Yabutani (yabutani@chem.tokushima-u.ac.jp)
Afework Kassu (afeworkkassu@yahoo.com)

**Version:** 2  **Date:** 7 January 2014

**Author's response to reviews:**

There are only topographic revisions made to this version of the manuscript in line with the BMC public health.